%  LaTeX support: latex@mdpi.com
%  For support, please attach all files needed for compiling as well as the log file, and specify your operating system, LaTeX version, and LaTeX editor.

%=================================================================
\documentclass[journal,article,submit,pdftex,moreauthors]{Definitions/mdpi}

%--------------------
% Class Options:
%--------------------
%----------
% journal
%----------
% Choose between the following MDPI journals:
% acoustics, actuators, addictions, admsci, adolescents, aerobiology, aerospace, agriculture, agriengineering, agrochemicals, agronomy, ai, air, algorithms, allergies, alloys, analytica, analytics, anatomia, animals, antibiotics, antibodies, antioxidants, applbiosci, appliedchem, appliedmath, applmech, applmicrobiol, applnano, applsci, aquacj, architecture, arm, arthropoda, arts, asc, asi, astronomy, atmosphere, atoms, audiolres, automation, axioms, bacteria, batteries, bdcc, behavsci, beverages, biochem, bioengineering, biologics, biology, biomass, biomechanics, biomed, biomedicines, biomedinformatics, biomimetics, biomolecules, biophysica, biosensors, biotech, birds, bloods, blsf, brainsci, breath, buildings, businesses, cancers, carbon, cardiogenetics, catalysts, cells, ceramics, challenges, chemengineering, chemistry, chemosensors, chemproc, children, chips, cimb, civileng, cleantechnol, climate, clinpract, clockssleep, cmd, coasts, coatings, colloids, colorants, commodities, compounds, computation, computers, condensedmatter, conservation, constrmater, cosmetics, covid, crops, cryptography, crystals, csmf, ctn, curroncol, cyber, dairy, data, ddc, dentistry, dermato, dermatopathology, designs, devices, diabetology, diagnostics, dietetics, digital, disabilities, diseases, diversity, dna, drones, dynamics, earth, ebj, ecologies, econometrics, economies, education, ejihpe, electricity, electrochem, electronicmat, electronics, encyclopedia, endocrines, energies, eng, engproc, entomology, entropy, environments, environsciproc, epidemiologia, epigenomes, est, fermentation, fibers, fintech, fire, fishes, fluids, foods, forecasting, forensicsci, forests, foundations, fractalfract, fuels, future, futureinternet, futurepharmacol, futurephys, futuretransp, galaxies, games, gases, gastroent, gastrointestdisord, gels, genealogy, genes, geographies, geohazards, geomatics, geosciences, geotechnics, geriatrics, grasses, gucdd, hazardousmatters, healthcare, hearts, hemato, hematolrep, heritage, higheredu, highthroughput, histories, horticulturae, hospitals, humanities, humans, hydrobiology, hydrogen, hydrology, hygiene, idr, ijerph, ijfs, ijgi, ijms, ijns, ijpb, ijtm, ijtpp, ime, immuno, informatics, information, infrastructures, inorganics, insects, instruments, inventions, iot, j, jal, jcdd, jcm, jcp, jcs, jcto, jdb, jeta, jfb, jfmk, jimaging, jintelligence, jlpea, jmmp, jmp, jmse, jne, jnt, jof, joitmc, jor, journalmedia, jox, jpm, jrfm, jsan, jtaer, jvd, jzbg, kidneydial, kinasesphosphatases, knowledge, land, languages, laws, life, liquids, literature, livers, logics, logistics, lubricants, lymphatics, machines, macromol, magnetism, magnetochemistry, make, marinedrugs, materials, materproc, mathematics, mca, measurements, medicina, medicines, medsci, membranes, merits, metabolites, metals, meteorology, methane, metrology, micro, microarrays, microbiolres, micromachines, microorganisms, microplastics, minerals, mining, modelling, molbank, molecules, mps, msf, mti, muscles, nanoenergyadv, nanomanufacturing,\gdef\@continuouspages{yes}} nanomaterials, ncrna, ndt, network, neuroglia, neurolint, neurosci, nitrogen, notspecified, %%nri, nursrep, nutraceuticals, nutrients, obesities, oceans, ohbm, onco, %oncopathology, optics, oral, organics, organoids, osteology, oxygen, parasites, parasitologia, particles, pathogens, pathophysiology, pediatrrep, pharmaceuticals, pharmaceutics, pharmacoepidemiology,\gdef\@ISSN{2813-0618}\gdef\@continuous pharmacy, philosophies, photochem, photonics, phycology, physchem, physics, physiologia, plants, plasma, platforms, pollutants, polymers, polysaccharides, poultry, powders, preprints, proceedings, processes, prosthesis, proteomes, psf, psych, psychiatryint, psychoactives, publications, quantumrep, quaternary, qubs, radiation, reactions, receptors, recycling, regeneration, religions, remotesensing, reports, reprodmed, resources, rheumato, risks, robotics, ruminants, safety, sci, scipharm, sclerosis, seeds, sensors, separations, sexes, signals, sinusitis, skins, smartcities, sna, societies, socsci, software, soilsystems, solar, solids, spectroscj, sports, standards, stats, std, stresses, surfaces, surgeries, suschem, sustainability, symmetry, synbio, systems, targets, taxonomy, technologies, telecom, test, textiles, thalassrep, thermo, tomography, tourismhosp, toxics, toxins, transplantology, transportation, traumacare, traumas, tropicalmed, universe, urbansci, uro, vaccines, vehicles, venereology, vetsci, vibration, virtualworlds, viruses, vision, waste, water, wem, wevj, wind, women, world, youth, zoonoticdis
% For posting an early version of this manuscript as a preprint, you may use "preprints" as the journal. Changing "submit" to "accept" before posting will remove line numbers.

%---------
% article
%---------
% The default type of manuscript is "article", but can be replaced by:
% abstract, addendum, article, book, bookreview, briefreport, casereport, comment, commentary, communication, conferenceproceedings, correction, conferencereport, entry, expressionofconcern, extendedabstract, datadescriptor, editorial, essay, erratum, hypothesis, interestingimage, obituary, opinion, projectreport, reply, retraction, review, perspective, protocol, shortnote, studyprotocol, systematicreview, supfile, technicalnote, viewpoint, guidelines, registeredreport, tutorial
% supfile = supplementary materials

%----------
% submit
%----------
% The class option "submit" will be changed to "accept" by the Editorial Office when the paper is accepted. This will only make changes to the frontpage (e.g., the logo of the journal will get visible), the headings, and the copyright information. Also, line numbering will be removed. Journal info and pagination for accepted papers will also be assigned by the Editorial Office.

%------------------
% moreauthors
%------------------
% If there is only one author the class option oneauthor should be used. Otherwise use the class option moreauthors.

%---------
% pdftex
%---------
% The option pdftex is for use with pdfLaTeX. Remove "pdftex" for (1) compiling with LaTeX & dvi2pdf (if eps figures are used) or for (2) compiling with XeLaTeX.

%=================================================================
% MDPI internal commands - do not modify
\firstpage{1}
\makeatletter
\setcounter{page}{\@firstpage}
\makeatother
\pubvolume{1}
\issuenum{1}
\articlenumber{0}
\pubyear{2024}
\copyrightyear{2024}
%\externaleditor{Academic Editor: Firstname Lastname}
\datereceived{ }
\daterevised{ } % Comment out if no revised date
\dateaccepted{ }
\datepublished{ }
%\datecorrected{} % For corrected papers: "Corrected: XXX" date in the original paper.
%\dateretracted{} % For corrected papers: "Retracted: XXX" date in the original paper.
\hreflink{https://doi.org/} % If needed use \linebreak
%\doinum{}
%\pdfoutput=1 % Uncommented for upload to arXiv.org
%\CorrStatement{yes}  % For updates


%=================================================================
% Add packages and commands here. The following packages are loaded in our class file: fontenc, inputenc, calc, indentfirst, fancyhdr, graphicx, epstopdf, lastpage, ifthen, float, amsmath, amssymb, lineno, setspace, enumitem, mathpazo, booktabs, titlesec, etoolbox, tabto, xcolor, colortbl, soul, multirow, microtype, tikz, totcount, changepage, attrib, upgreek, array, tabularx, pbox, ragged2e, tocloft, marginnote, marginfix, enotez, amsthm, natbib, hyperref, cleveref, scrextend, url, geometry, newfloat, caption, draftwatermark, seqsplit
% cleveref: load \crefname definitions after \begin{document}
\begin{thebibliography}{99}
\bibitem{gu} 1
Gunawardena ~J. \emph{Idempotency}. Publ. Newton Inst., vol. 11,
Cambridge University Press, Cambridge, 1998.

\bibitem{Green}2
Cuninghame-Green, R.A. \emph{Minimax Algebra}, Lecture Notes in
Economics and Mathematical Systems, vol.166, Springer Verlag, 1979.

\bibitem{Kolokoltsov-Maslow}3
Kolokoltsov, V.N.; Maslov, V.P. \emph{Idempotent Analysis and its
Applications}, Springer, 1997.

\bibitem{rt1}4
Radul ~T. Idempotent measures: absolute retracts and soft mappings.
\emph{Topological Methods in Nonlinear Analysis }[online] {\bf
2020}, \emph{56}, 161--172.

\bibitem{za2}5
Zaitov ~A.A. On a metric on the space of idempotent probability
measures. \emph{Appl. Gen. Topol.} {\bf 2020}, \emph{16}, 35--51.

\bibitem{zi}6
Zaitov ~A.A..; Ishmetov ~A.Ya. Homotopy properties of the space
$I_{f}(X)$ of idempotent probability measures. \emph{Math. Notes}
{\bf 2019}, \emph{106}, 562--571.

\bibitem{zm}7
Zarichnyi ~M. Spaces and mappingpings of idempotent measures.
\emph{Izve. Math.} {\bf 2010}, \emph{74}, 45--64.

\bibitem{zbr}8
Zarichnyi ~M., Bazylevych ~L., Repovs ~D. Spaces of idempotent
measures of compact metric spaces. \emph{Topol. Appl.} {\bf 2010},
\emph{157}, 136--144.

\bibitem{ba}9
Banakh ~T.O. Topology of probability measures spaces, I: The
functors $P_{\tau}$ and $\hat{P}$. \emph{Mat. Stud.} {\bf1995},
\emph{5}, 65--87.

\bibitem{br}10
Banakh T.O.; Radul T.N. Topology of spaces of probability measures.
\emph{Sb. Math.} {\bf 1997}, \emph{188}, 973--995.

\bibitem{fedorchuk}11
Fedorchuk, V.V. Probability measures in topology, \emph{Uspekhi Mat.
Nauk} {\bf 1991}, \emph{46},  41--80; English transl. \emph{Russian
Math. Surveys} {\bf 1991}, \emph{46},  45--93.

\bibitem{aaz}12
Albeverio ~S.; Ayupov ~Sh.A.; Zaitov ~A.A. On certain properties of
the spaces of order-preserving functionals. \emph{Topol. Appl.} {\bf
2008}, \emph{155}, 1792--1799.

\bibitem{az}13
Ayupov ~Sh.A.; Zaitov ~A.A. Functor of weakly additive $\tau$-smooth
functionals and mappingpings. \emph{Ukrainian Math. J.} {\bf 2009},
\emph{61}, 1380--1386.

\bibitem{za1}14
Zaitov ~A.A. The functor of order-preserving functionals of finite
degree. \emph{J. Math. Sci.} {\bf 2006}, \emph{133}, 1602--1603.

\bibitem{rt}15
Radul ~T. On the functor of order-preserving functionals,
\emph{Comment. Math. Univ. Carolin.} {\bf 1998}, \emph{39},
609--615.

\bibitem{zk}16
Zaitov ~A.A., Kholturayev ~Kh.F. On some properties of infinite
iterations of the functor of idempotent probobility measures.
\emph{Lobachevskii J. Math.} {\bf 2022}, \emph{43}, 2341--2348.

\bibitem{fs}17
Fedorchuk ~V.V., Sadovnichii ~Yu.V. On some topological and
categorical properties of measures with alternating signs.
\emph{Fundam. Prikl. Mat.} {\bf 1999}, \emph{23}, 597--618.

\bibitem{va}18
Varadarajan, ~V.~S. Measures on topological spaces, \emph{Mat. Sb.
(N.S.)} {\bf 1961}, \emph{55},  35--100 (in Russian).

\bibitem{pu} 19
Puhalskii ~A. \emph{Large Deviations and Idempotent Probability}.
Chapman \& Hall/CRC, vol. 500, 2001.

\bibitem{es} 20
Eshimbetov ~M.R. Luzin measurable functions and idempotent
$\tau$-smooth measures.\emph{ Bull. Inst. Math. Uzbek Acad. Sci.}
{\bf 2021}, \emph{7}, 136--142.

\bibitem{es1}21
Eshimbetov ~M.R. On extensions and restrictions of $\tau$-maxitive
idempotent measures. \emph{Uzbek Math. J.} {\bf 2021}, \emph{65},
71--80.

\bibitem {es2}22
Eshimbetov ~M.R. On extensions and restrictions of $\tau$-smooth and
$\tau$-maxitive idempotent measures. \emph{Bull. Nat. Univ.
Uzbekistan: Math.  Nat. Sci.} {\bf 2021}, \emph{4}, 3--15.

\bibitem{er}23
Engelking ~R. \emph{General Topology}. Revised and completed ed.
Heldermann, Berlin, Germany, 1989.

\end{thebibliography}


%=================================================================
% Please use the following mathematics environments: Theorem, Lemma, Corollary, Proposition, Characterization, Property, Problem, Example, ExamplesandDefinitions, Hypothesis, Remark, Definition, Notation, Assumption
%% For proofs, please use the proof environment (the amsthm package is loaded by the MDPI class).

%=================================================================
% Full title of the paper (Capitalized)
\Title{On \v{C}ech-completeness of the space of $\tau$-smooth
idempotent probability measures}

% MDPI internal command: Title for citation in the left column
\TitleCitation{On \v{C}ech-completeness of $I_{\tahu}(X)$}

% Author Orchid ID: enter ID or remove command
\newcommand{\orcidauthorA}{0000-0000-0000-000X} % Add \orcidA{} behind the author's name
%\newcommand{\orcidauthorB}{0000-0000-0000-000X} % Add \orcidB{} behind the author's name

% Authors, for the paper (add full first names)
\Author{Lj. D. R. Ko\v cinac
$^{1,\dagger,\ddagger}$\orcid{000-0-0002-4870-7908}}, A. A. Zaitov
$^{2}$ and M. R. Eshimbetov $^{3}$*}

%\longauthorlist{yes}

% MDPI internal command: Authors, for metadata in PDF
\AuthorNames{Lj. D. R. Ko\v cinac, A. A. Zaitov, and M. R.
Eshimbetov}

% MDPI internal command: Authors, for citation in the left column
\AuthorCitation{Ko\v cinac, Lj.D.R.; Zaitov, Eshimbetov, M.R.}
% If this is a Chicago style journal: Lastname, Firstname, Firstname Lastname, and Firstname Lastname.

% Affiliations / Addresses (Add [1] after \address if there is only one affiliation.)
\address{%
$^{1}$ \quad University of Ni\v s, Faculty of Sciences and Mathematics, 18000 Nis, Serbia; lkocinac2gmail.com\\
$^{1}$ \quad State University of Noni Pazar, 36230 Novj Pazar,
Serbia\\
$^{2}$ \quad Tashkent Architecture and Civil Engineering University,
Yangi Shahar Str. 9, 100194 Tashkent, Uzbekistan;
adilbek\_zaitov@mail.ru\\
$^{3}$ \quad Institute of Mathematics named after V.~I.~Romanovsky,
Universitet Str. 9, 100174 Tashkent, Uzbekistan;
mr.eshimbetov@gmail.com }



% Contact information of the corresponding author
\corres{Correspondence: lkocinac@gmail.come-mail.com}}

% Current address and/or shared authorship
%\firstnote{Current address: Affiliation 3.}
\secondnote{These authors contributed equally to this work.}
% The commands \thirdnote{} till \eighthnote{} are available for further notes

%\simplesumm{} % Simple summary

%\conference{} % An extended version of a conference paper

% Abstract (Do not insert blank lines, i.e. \\)
\abstract{For the set of probability measures ${\sf I}(X)$, where
$X$ is a compact Hausdorff space, we propose a new way to introduce
the topology by using open subsets of the space $X$. Then, among
other things,  we give a new proof that for a compact Hausdorff
space $X$ the space ${\sf I}(X)$ is also a compact Hausdorff space.
For a Tychonoff space $X$, we consider the topological space ${\sf
I_{\tau}}(X)$ of $\tau$-smooth idempotent probability measures on
$X$,and show that the space ${\sf I_{\tau}}(X)$ is \v{C}ech-complete
if and only if the given space $X$ is \v{C}ech-complete. }

% Keywords
\keyword{\v{C}ech-complete space, compact space, probability
measure, $\tau$-smooth idempotent probability measure, neighbourhood
system.}

% The fields PACS, MSC, and JEL may be left empty or commented out if not applicable
%\PACS{J0101}
\MSC{28C15, 28A33, 46E27, 60B05, 54D99}
%\JEL{}

\section{Introduction}

Idempotent mathematics is a branch of mathematical sciences, rapidly
developing and gaining popularity over the last four decades. An
important stage of development of the subject was presented in the
book ``Idempotency'' \cite{gu} edited by J.~Gunawardena (see also
\cite{Green, Kolokoltsov-Maslow}). This book arose out of the
well-known international workshop that was held in Bristol, England,
in October 1994. Idempotent mathematics is based on replacing the
usual arithmetic operations with a new set of basic operations,
i.~e., on replacing numerical fields by idempotent semirings and
semifields. Typical example is the so-called max-plus algebra (in
fact, an idempotent semifield) $\mathbb{R}_{\max}= \mathbb R \cup
\{-\infty\}$ with operation  $x \oplus y = \max\{x,y\}$ and $x\odot
y = x+y$ (\cite{gu}, \cite{rt1}, \cite{va}, \cite{za1}).

M.~Zarichnyi in 2010 in \cite{zm} investigated categorical
properties of the space of idempotent probability measures. In
\cite{zbr} the theory was extended to the class of metric spaces.
So, the space of idempotent probability measures is a new object.
But it is already studied from different points of view in Measure
Theory, Functional Analysis, Probability Theory, Topology and
Category Theory. The study of spaces of idempotent probability
measures leads to the problem its investigations on wider classes of
topological classes than the class of compact Hausdorff spaces, in
particular, the class of Tychonoff spaces.

T.~Banakh \cite{ba}, T.~Banakh and T.~Radul \cite{br},
\cite{fedorchuk}, carried out a systematic study on probability
measures on Tychonoff spaces. In their studies, they fruitfully used
the linearity of probability measures. Unlike probability measures,
idempotent probability measures are not linear. In papers
\cite{aaz}, \cite{az} and \cite{za1} the theory put forward, and in
\cite{rt} some categorical properties of $\tau$-smooth weakly
additive (nonlinear) functionals were established.

The results obtained in  \cite{rt1}, \cite{va}, \cite{za1},
\cite{zk}, \cite{zm}, \cite{zbr} show that in order to establish
"good" properties of the space of idempotent probability measures,
methods are required that are very different from classical methods
(i.~e., from methods suitable for probability measures which have
been productively used in \cite{fs}, \cite{va} and others).

Unlike above mentioned papers \cite{rt1}, \cite{va}, \cite{za1},
\cite{zk}, \cite{zm}, \cite{zbr}, in this paper for a compact
Hausdorff space $X$ we introduce the notion of idempotent measures
as a set-function on the family $\mathfrak{B}(X)$. Note that the
works \cite{pu}, \cite{es}, \cite{es1} and \cite{es2} also have this
approach. Improving their theory in the current paper we note some
types of open and closed subsets of the space of idempotent
probability measures. In the set of idempotent probability measures,
we introduce the base of the product topology and show that for a
compact Hausdorff space $X$ the topological space ${\sf I}(X)$ of
idempotent probability measures is also a compact Hausdorff space.

Further, for a  Tychonoff space $X$, we consider the space
$I_{\tau}(X)$ of $\tau$-smooth idempotent probability measures on
$X$. Then we establish the \v{C}ech-completeness of the space of
$\tau$-smooth idempotent probability measures for the
\v{C}ech-complete Tychonoff space. (\v{C}ech-complete spaces were
introduced by Eduard \v{C}ech in 1937 to prove the Baire category
theorem. Another important application of the \v{C}ech-completeness
appears in the metrization of a topological space by a complete
metric. Note that a topological space $X$ is \v{C}ech-complete if
$X$ is a Tychonoff space and the remainder $\beta X\setminus X$ is
an $F_{\sigma}$-set in the Stone-\v{C}ech compactification $\beta
X$. Locally compact spaces are \v{C}ech-complete, but the inverse is
not take place. The space of all irrational numbers with the
topology of a subspace of the real line is an example of a
\v{C}ech-complete space that is not locally compact \cite{er}.) From
here, since \v{C}ech-completeness is hereditary with respect to
$G_{\delta}$-subsets and a \v{C}ech-complete Tychonoff space $X$ is
$G_{\delta}$-subset in its Stone-\v{C}ech compactification $\beta
X$, we conclude that a Tychonoff space $X$ is \v{C}ech-complete if
and only if $I_{\tau}(X)$ is \v{C}ech-complete. Note that a linear
(in the classical sense) version of this result was established in
\cite{ba}. One can see that the methods used in \cite{aaz},
\cite{az}, \cite{ba}, \cite{br}, \cite{fs}, \cite{va}, \cite{za1}
are not suitable for the present case.

\section{Preliminaries}

Let $X$ be a compact Hausdorff space and $\mathfrak{B}(X)$ the
family of Borel subsets of $X$. We denote $\overline{\mathbb{R}}_{+}
=\left[0,\, +\infty \right)\cup \left\{+\infty \right\}=\left[0,\,
+\infty \right]$.  The symbol $D$ denotes the directed set.
Following \cite{pu}, we enter the following notion.


\begin{Definition}\rm \label{d1}
A set function $\mu\colon\, \mathfrak{B}(X)\to
\overline{\mathbb{R}}_{+}$ is said to be  an \emph{idempotent
measure} on $X$ if the following conditions hold:
\begin{enumerate}
\item[$1)$] $\mu (\emptyset) = 0$;

\item[$2)$] $\mu (A\cup B) = \max \{\mu (A),\, \mu (B)\}$ for any $A,\, B\in \mathfrak{B}(X)$;

\item[$3)$] $\mu \left(\underset{\alpha\in D}{\bigcup}A_{\alpha}\right)
= \mathop{\sup}\limits_{\alpha\in D} \{\mu (A_{\alpha})\}$ for every
increasing net $\{A_{\alpha},\, \alpha\in D\}
\subset\mathfrak{B}(X)$ such that $\underset{\alpha \in
D}{\cup}A_{\alpha} \in \mathfrak{B}(X)$.
\end{enumerate}
\end{Definition}

\begin{Remark} \label{r1}{\rm
Every idempotent measure $\mu$ is increasing, i.~e. for $A$, $B$
$\in \mathfrak{B}(X)$ if $A\subset B$, then $\mu(A)\leq\mu(B)$.}
\end{Remark}

The set of all idempotent measure on $X$ we denote by $IM(X)$. If
$\mu(X)=1$, the idempotent measure $\mu$ is called an
\emph{idempotent probability measure} on $X$. We denote
\begin{gather*}
{\sf I}(X) = \{\mu\in IM(X)\colon\, \mu(X)=1\}.
\end{gather*}


A set
\begin{gather}\label{supp1}
\operatorname{supp}\,\mu = \left[X\setminus \cup\{A\in
\mathfrak{B}(X):\, \mu(A) = 0\}\right]_{X}
\end{gather}
is said to be a \textit{support} of the given idempotent measure
$\mu$.

The support of an idempotent measure $\mu$ can be defined by the
following equality:
\begin{gather}\label{supp2}
\operatorname{supp}\,\mu = \left[\cap\{C\colon\, C\in
\mathfrak{B}(X),\, \mu(X\setminus C) = 0\}\right]_{X}.
\end{gather}


\begin{Lemma}\label{l2}
For every $\mu\in {\sf I}(X)$ we have $\mu\in
I(\operatorname{supp}\,\mu)$.
\end{Lemma}

\begin{proof} For a support we will apply (\ref{supp2}). It is easy to see that
\begin{align*}
\mu(X\setminus \operatorname{supp}\,\mu) &= \mu\left(X\setminus\left[\cap\{C\colon\, C\in \mathfrak{B}(X),\, \mu(X\setminus C)=0\}\right]_{X}\right)\\
&=\mu\left(\operatorname{int}\left(X\setminus\cap\{C\colon\, C\in \mathfrak{B}(X),\, \mu(X\setminus C)=0\}\right)\right)\\
&= \mu\left(\operatorname{int}\left(\cup\{X\setminus C\colon\, C\in \mathfrak{B}(X),\, \mu(X\setminus C)=0\}\right)\right)\\
&\leq \mu\left(\cup\{X\setminus C\colon\, C\in \mathfrak{B}(X),\, \mu(X\setminus C)=0\}\right)\\
&= \mathop{\sup}\{\mu(X\setminus C):\, C\in \mathfrak{B}(X),\,
\mu(X\setminus C)=0\}=0.
\end{align*}
On the other side,
\begin{gather*}
1 = \mu(X)=\mu\left(\left(X\setminus
\operatorname{supp}\,\mu\right)\cup
\operatorname{supp}\,\mu\right)=\mathop{\max}\{\mu(X\setminus
\operatorname{supp}\,\mu),\, \mu(\operatorname{supp}\,\mu)\},
\end{gather*}
i.~e. $\mu(\operatorname{supp}\,\mu)=1$ and $\mu\in
I(\operatorname{supp}\,\mu)$.
\end{proof}


Let $X$ be a compact Hausdorff space, $\mathcal{B}$ a base in $X$,
$U_{i}\in \mathcal{B}$, $i=1,\, \ldots, \, n$, and $\varepsilon >
0$. For an idempotent probability measure $\mu\in {\sf I}(X)$ we
define a set
$$
\langle\mu;\, U_1,\,\ldots,\, U_n;\, \varepsilon\rangle = \{\nu\in
{\sf I}(X)\colon\, |\nu(U_i) - \mu(U_i)| <\varepsilon,\, i=1,\,
\ldots, \, n\}.
$$

Gathering all of such sets construct a family
$$
\mathcal{B}(\mu) = \{\langle\mu;\, U_1,\,\ldots,\, U_n;\,
\varepsilon\rangle\colon\, U_{i}\in \mathcal{B},\, i=1,\, \ldots, \,
n; \varepsilon > 0\}, \quad \mu\in I(X),
$$
and put
$$
\mathcal{B}_{{\sf I}(X)} = \underset{\mu\in {\sf
I}(X)}{\cup}\mathcal{B}(\mu).
$$

\begin{Proposition}\label{p1}
The built family $\mathcal{B}_{{\sf I}(X)}$ forms a base (or, a
neighbourhoods system) for some topology in ${\sf I}(X)$.
\end{Proposition}

\begin{proof}
Since $\mathcal{B} \ne \emptyset$, there exists an open nonempty set
$U\in \mathcal{B}$. On the other side, $U\in \mathfrak{B}(X)$, so
$\mu(U)$ is well defined. That is why for every $\varepsilon > 0$ we
have $\langle\mu;\, U;\, \varepsilon\rangle \in \mathcal{B}(\mu)$.
Consequently, $\mathcal{B}(\mu) \ne \emptyset$. Clearly, $\mu\in
\langle\mu;\, U_1,\,\ldots,\, U_n;\, \varepsilon\rangle$ for each
$\langle\mu;\, U_1,\,\ldots,\, U_n;\, \varepsilon\rangle \in
\mathcal{B}(\mu)$.

Suppose  we get $\mu\in \langle\nu;\, V_1,\,\ldots,\, V_n;\,
\delta\rangle\in \mathcal{B}(\nu)$. Note that $(\mu - \nu)(V) =
\mu(V) - \nu(V)$, and designate
\begin{gather*}
a=\min\{(\nu - \mu)(V_{1}) + \delta,\, (\mu - \nu)(V_{1}) +
\delta,\,\ldots,\, (\nu - \mu)(V_{n}) + \delta,\, (\mu - \nu)(V_{n})
+ \delta\}.
\end{gather*}
Obviously, $a > 0$. Then for every $\xi\in \langle\mu;\,
V_1,\,\ldots,\, V_n;\, a\rangle$ one has
\begin{align*}
|\xi(V_{i}) - \nu(V_{i})| &= |\xi(V_{i}) - \mu(V_{i}) + \mu(V_{i}) - \nu(V_{i})|\leq\\
&\leq |\xi(V_{i}) - \mu(V_{i})| + |\mu(V_{i}) - \nu(V_{i})|<\\
&< a + |\mu(V_{i}) - \nu(V_{i})|.
\end{align*}
Two cases are possible: $\mu(V_{i}) - \nu(V_{i}) \geq 0$ or $\mu(V_{i}) - \nu(V_{i}) \leq 0$. In the front case we take into attention $a < \nu(V_{i}) - \mu(V_{i}) + \delta$, and the last case:  $a < \mu(V_{i}) - \nu(V_{i}) + \delta$. Hence, $\xi\in \langle\nu;\, V_1,\,\ldots,\, V_n;\, \delta\rangle$, in the other words, $\langle\mu;\, V_1,\,\ldots,\, V_n;\, a\rangle\subset \langle\nu;\, V_1,\,\ldots,\, V_n;\, \delta\rangle$.\\


Finally, consider an arbitrary couple of sets $\langle\mu;\,
U_1,\,\ldots,\, U_n;\, \varepsilon\rangle$, $\langle\mu;\,
V_1,\,\ldots,\, V_k;\, \delta\rangle$ belonging $\mathcal{B}(\mu)$.
Denote $\theta = \min\{\varepsilon,\, \delta\}$, and we have
\begin{align*}
&\langle\mu;\, U_1,\,\ldots,\, U_n;\, \varepsilon\rangle\cap \langle\mu;\, V_1,\,\ldots,\, V_k;\, \delta\rangle \supset\\
\supset &\langle\mu;\, U_1,\,\ldots,\, U_n;\, \theta\rangle\cap \langle\mu;\, V_1,\,\ldots,\, V_k;\, \theta\rangle = \\
= &\langle\mu;\, U_1,\,\ldots,\, U_n,\, V_1,\,\ldots,\, V_k;\,
\theta\rangle\in \mathcal{B}(\mu).
\end{align*}
Proposition \ref{p1} is proved.
\end{proof}

\begin{Remark}\label{r2} {\rm For an open set $U$ sets
\begin{gather*}
\langle U;\, \varepsilon\rangle_{-} = \{\xi\in {\sf I}(X)\colon\,
\xi(U) < \varepsilon\}\,\, \mbox{ and }\,\,  \langle U;\,
\varepsilon\rangle_{+} = \{\xi\in I(X)\colon\, \xi(U) >
\varepsilon\}
\end{gather*}
are open. Indeed, we have $\mu(U) = a < \varepsilon$ for every
$\mu\in \langle U;\, \varepsilon\rangle_{-}$. Then for each $\nu\in
\langle \mu;\, U;\, \varepsilon - a \rangle_{-}$ one has
$-(\varepsilon - a) + \mu(U) < \nu(U) < \mu(U) + (\varepsilon - a)$,
$2a-\varepsilon < \nu(U) < \varepsilon$. Hence, $\nu\in\langle U;\,
\varepsilon\rangle_{-}$. The openness of the second set will be
established like to the openness of the first set.}
\end{Remark}

\begin{Remark}\label{r3} {\rm For a closed set $F$ sets
\begin{gather*}
[ F;\, \varepsilon]_{-} = \{\xi\in {\sf I}(X)\colon\, \xi(F) \leq
\varepsilon\}\,\, \mbox{ and }\,\, [ F;\, \varepsilon]_{+} =
\{\xi\in I(X)\colon\, \xi(F) \geq \varepsilon\}
\end{gather*}
are closed. Indeed, for each $\{\xi_{\alpha}\} \subset [ F;\,
\varepsilon]_{-} \subset {\sf I}(X)$ there exists $\xi_{0}\in {\sf
I}(X)$ such that $\xi_{0} = \underset\alpha\lim\, \xi_{\alpha}$
(i.~e, $\xi(A) = \underset\alpha\lim\, \xi_{\alpha}(A)$ for each
$A\in \mathfrak{B}(X)$). Hence, $\xi_{0}(F) \leq \varepsilon$. Thus,
$\xi_{0}\in[ F;\, \varepsilon]_{-}$. In the same way one can show
that $[F;\, \varepsilon]_{+}$ is also closed.}
\end{Remark}

\section{Idempotent probability measures on compact Hausdorff spaces}

The construction of $I(X)$ gives that ${\sf I}(X)\subset
\mathbb{R}^{\mathfrak{B}(X)}$. Equip the set ${\sf I}(X)$ with the
topology generated by the above neighbourhoods system. Obviously,
this topology coincides with the induced topology from the product
topology of $\mathbb{R}^{\mathfrak{B}(X)}$ to ${\sf I}(X)$.

\begin{Theorem}\label{t1}
For a compact Hausdorff space $X$ the topological space ${\sf I}(X)$
is also a compact Hausdorff space.
\end{Theorem}
\begin{proof}
At first we show that ${\sf I|}(X)$ is a Hausdorff space. Let $\mu$,
$\nu\in {\sf I}(X)$ be different idempotent measures. Then there
exists a set $U\in\mathcal{B}$ such that $\mu(U) \neq \nu(U)$. Put
$|\mu(U) - \nu(U)| = a$. One has $a\ne 0$, $|\mu(U) - \nu(U)| >
\tfrac{a}{2}$. Assume that there exists an idempotent probability
measure $\xi$ belonging to both sets $\langle\mu;\, U;\,
\tfrac{a}{4}\rangle$ and $\langle\nu;\, U;\, \tfrac{a}{4}\rangle$.
Then
\begin{align*}
|\nu(U) - \mu(U)| &= |\nu(U) - \xi(U) + \xi(U) - \mu(U)| \leq\\
& \leq|\nu(U) - \xi(U)| + |\xi(U) - \mu(U)| < \tfrac{a}{4} +
\tfrac{a}{4} = \tfrac{a}{2}.
\end{align*}
The obtained contradiction shows that $\langle\mu;\, U;\,
\tfrac{a}{4}\rangle \cap \langle\nu;\, U;\, \tfrac{a}{4}\rangle
=\emptyset$.


Now we will prove the compactness of $I(X)$ with respect to the
topology generated by the  neighbourhoods system $\mathcal{B}_{{\sf
I}(X)}$.

Clearly, ${\sf I}(X)$ is bounded in $\mathbb{R}^{\mathfrak{B}(X)}$.
It remains to show its closedness. Let $\mu\in
\mathbb{R}^{\mathfrak{B}(X)}\setminus I(X)$. Then the following
cases are possible.

\emph{Case 1}. $\mu(\emptyset) = a \ne 0$. In this case we have
$\langle \mu;\, \emptyset;\, \tfrac{|a|}{2}\rangle \cap {\sf I}(X) =
\emptyset$.

\emph{Case 2}. $\mu(X) = a \ne 1$. Then it os obvious that $\langle
\mu;\, X;\, \tfrac{|a+1|}{2}\rangle \cap {\sf I}(X) = \emptyset$.
\\

\emph{Case 3}. $\mu(G\cup \Gamma) = a \ne b = \max\{\mu(G),\,
\mu(\Gamma)\}$ for some open sets $G$ and $\Gamma$. Whereupon,
$\langle \mu;\, G,\, \Gamma,\, G\cup \Gamma;\, \tfrac{|b -
a|}{2}\rangle \cap I(X) = \emptyset$.

\emph{Case 4}. $\mu\left(\underset{\alpha\in D}{\bigcup}
\Gamma_{\alpha}\right) = a \ne b = \underset{\alpha\in
D}{\sup}\{\mu(\Gamma_{\alpha})\}$ for a net
$\{\Gamma_{\alpha}\}_{\alpha\in D}$ of open sets.

The fourth case has the following subcases.

\emph{Subcase 4.1}. Assume $0\leq a < b \leq 1$. By the definition
there exist such an index $\alpha_{0}$ that
$\mu(\Gamma_{\alpha_{0}}) > b - \tfrac{b - a}{4} = \tfrac{a +
3b}{4}$. We claim that $\langle \mu;\, \Gamma_{\alpha_{0}},\,
\underset{\alpha\in D}{\cup} \Gamma_{\alpha};\, \tfrac{b -
a}{4}\rangle \cap {\sf I}(X) = \emptyset$. Presume, it is not so.
Then there exists $\nu$ belonging to this intersection. From here we
get
\begin{align*}
\left|\nu\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right) - \mu\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right)\right| &< \tfrac{b - a}{4},\\
|\nu(\Gamma_{\alpha_{0}}) - \mu(\Gamma_{\alpha_{0}})| &< \tfrac{b -
a}{4}.
\end{align*}
These inequalities provide, correspondingly,
\begin{gather*}
\tfrac{3a - b}{4} < \nu\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right) < \tfrac{3a + b}{4},\\
\tfrac{a + b}{2} < \nu(\Gamma_{\alpha_{0}}) < \tfrac{5b - a}{4}.
\end{gather*}
In the issue one obtains
\begin{gather*}
\underset{\alpha\in D}{\sup}\{\nu(\Gamma_{\alpha})\} \geq
\nu(\Gamma_{\alpha_{0}}) > \tfrac{a + b}{2} > \tfrac{3a + b}{4} >
\nu\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right),
\end{gather*}
i.e. $\underset{\alpha\in D}{\sup}\{\nu(\Gamma_{\alpha})\} >
\nu\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right)$.
The last inequality goes against $\nu\in {\sf I}(X)$. The resulting
contradiction shows that our assumption is wrong, and our statement
about the emptiness of the intersection is true.

\emph{Subcase 4.2}. Assume $0\leq b < a \leq 1$. Take an any
$\xi\in\langle \mu;\, \underset{\alpha\in D}{\cup}
\Gamma_{\alpha};\, \tfrac{a - b}{4}\rangle$. Then
\begin{gather*}
\underset{\alpha\in D}{\sup}\{\xi(\Gamma_{\alpha})\}  < b + \tfrac{a
- b}{4} < a - \tfrac{a - b}{4} = \mu\left(\underset{\alpha\in
D}{\bigcup} \Gamma_{\alpha}\right) - \tfrac{a - b}{4} <
\xi\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right),
\end{gather*}
i.~e. $\underset{\alpha\in D}{\sup}\{\xi(\Gamma_{\alpha})\} <
\xi\left(\underset{\alpha\in D}{\bigcup} \Gamma_{\alpha}\right)$.
Hence, $\xi\not \in I(X)$. Thus, $\langle \mu;\, \underset{\alpha\in
D}{\cup} \Gamma_{\alpha};\, \tfrac{a - b}{4}\rangle\cap {\sf I}(X) =
\emptyset$.

\emph{Subcase 4.3}. Assume $b < 0$. Then $\mu(\Gamma_{\alpha})\leq
b$ for every $\alpha\in D$. For $\tfrac{|b|}{2}$ there exists an
index $\alpha_{0}\in D$ such that $\frac{3b}{2} = b - \tfrac{|b|}{2}
< \mu(\Gamma_{\alpha_{0}}) < b + \tfrac{|b|}{2} = \frac{b}{2}$.
Evidently, $\langle \mu;\, \Gamma_{\alpha_{0}};\,
\tfrac{|b|}{2}\rangle \cap {\sf I}(X) = \emptyset$.

\emph{Subcase 4.4}. Assume $a < 0$. Then $\langle \mu;\,
\underset{\alpha\in D}{\cup} \Gamma_{\alpha};\,
\tfrac{|a|}{2}\rangle \cap {\sf I}(X) = \emptyset$.

\emph{Subcase 4.5}. Assume $b > 1$. Then there exists an index
$\alpha_{0}\in D$ such that $\tfrac{b + 1}{2} = b - \tfrac{b - 1}{2}
< \mu(\Gamma_{\alpha_{0}}) < b + \tfrac{b - 1}{2} = \tfrac{3b -
1}{2}$. Hence, $\langle \mu;\, \Gamma_{\alpha_{0}};\, \tfrac{b -
1}{2}\rangle \cap {\sf I|}(X) = \emptyset$.

\emph{Subcase 4.6}. Assume $a > 1$. Then $\langle \mu;\,
\underset{\alpha\in D}{\cup} \Gamma_{\alpha};\, \tfrac{a -
1}{2}\rangle \cap {\sf I}(X) = \emptyset$.

On this, all possible cases have been exhausted. Consequently, $\mu$
is an interior point of $\mathbb{R}^{\mathfrak{B}(X)}\setminus {\sf
I}(X)$ which immediately implies that ${\sf I}(X)$ is closed in
$\mathbb{R}^{\mathfrak{B}(X)}$ which means its compactness. The
proof of Theorem \ref{t1} is completed.
\end{proof}

For a mapping $f\colon\, X\to Y$ of compact Hausdorff spaces $X$ and
$Y$ we define a mapping
\begin{gather*}
{\sf I}(f)\colon\, {\sf I}(X)\to {\sf I}(Y)
\end{gather*}
by the rule
\begin{gather*}
{\sf I}(f)(\mu)(B) = \mu(f^{\gets}(B)),\qquad B\in \mathfrak{B}(Y).
\end{gather*}

\begin{Proposition}\label{p2}
For every pair of compact Hausdorff spaces $X$ and $Y$ and any
continuous mapping $f\colon\, X\to Y$ the mapping ${\sf
I}(f)\colon\, {\sf I}(X)\to {\sf I}(Y)$ is continuous.
\end{Proposition}

\begin{proof}
Take any $\mu\in {\sf I}(X)$, and let $\nu = {\sf I}(f)(\mu)\in {\sf
I}(Y)$. Consider an arbitrary neighbourhood $\langle\nu;\, V_{1},\,
\ldots,\, V_{k}; \delta\rangle$ of $\nu$. Then for any $\xi\in
\langle\mu;\, f^{\gets}(V_{1}),\, \ldots,\, f^{\gets}(V_{k});
\delta\rangle$ we have
\begin{gather*}
|{\sf I}(f)(\xi)(V_{i})) - \nu(V_{i}))| = |\xi(f^{\gets}(V_{i})) -
\mu(f^{\gets}(V_{i}))| < \delta,
\end{gather*}
which yields ${\sf I}(f)(\xi)\in \langle\nu;\, V_{1},\, \ldots,\,
V_{k}; \delta\rangle$. By virtue of arbitrariness of $\xi$ we obtain
that ${\sf I}(f)\left(\langle\mu;\, f^{\gets}(V_{1}),\, \ldots,\,
f^{\gets}(V_{k}); \delta\rangle\right)\subset  \langle\nu;\,
V_{1},\, \ldots,\, V_{k}; \delta\rangle$.
\end{proof}


\section{$\tau$-smooth idempotent probability measures}

To continue our investigation we need the notion of an outer
idempotent measure.

An \emph{outer idempotent measure} of an arbitrary subset $A\subset
X$ defines as
\begin{gather*}
\mu^{\ast}(A) = \mathop{\inf }\{\mu(B):\, B \supset A,\, B\in
\mathfrak{B}(X)\}.
\end{gather*}

So, we got an extension $\mu^{\ast}$ of $\mu$. Now Remark \ref{r1}
may be improved as follows

\begin{Remark} \label{r4}{\rm
For every idempotent measure $\mu$ its extension $\mu^{\ast}$ is
increasing, i.~e. if $A\subset B$, then
$\mu^{\ast}(A)\leq\mu^{\ast}(B)$.}
\end{Remark}

\begin{Lemma}\label{l3}
For any pair of $A$, $B\subset X$ and every idempotent measure we
have
\begin{gather*}
\mu^{\ast}(A\cup B)=\mathop{\max}\{\mu^{\ast}(A),\, \mu^{\ast}(B)\}.
\end{gather*}
\end{Lemma}

\begin{proof}
It is easy to see that $\mu^{\ast}(A\cup B)\geq
\mathop{\max}\{\mu^{\ast}(A),\, \mu^{\ast}(B)\}$. Let us show the
inverse inclusion, i.~e. $\mu^{\ast}(A\cup B)\leq
\mathop{\max}\{\mu^{\ast}(A),\, \mu^{\ast}(B)\}$. By the definition
of the outer idempotent measure one has
\[
\mu^{\ast}(A\cup B) = \mathop{\inf}\{\mu(E)\colon\, E\in
\mathfrak{B}(X),\, E\supset A\cup B\}.
\]
Clearly,
\[
\{E\colon\, E\in \mathfrak{B}(X),\, E\supset A\cup B\} \supset
\{C\cup D\colon\, C,\, D\in \mathfrak{B}(X),\, C\supset A\, \mbox{
and }\, D\supset B\}.
\]
Then
\begin{align*}
\mu^{\ast}(A\cup B) &= \mathop{\inf}\{\mu(E)\colon\, E\in \mathfrak{B}(X),\, E\supset A\cup B\}\leq\\
&\leq \mathop{\inf}\{\mu(C\cup D)\colon\, C,\, D\in \mathfrak{B}(X),\ C\supset A\, \mbox{ and }\, D\supset B\}=\\
&=\mathop{\inf}\{\mathop{\max}\{\mu(C),\, \mu(D)\}\colon\, C,\, D\in \mathfrak{B}(X),\ C\supset A\, \mbox{ and }\, D\supset B\}=\\
&=\mathop{\max}\left\{\mathop{\inf}\{\mu(C)\colon\, C\in \mathfrak{B}(X),\, C\supset A\},\ \mathop{\inf}\{\mu(D)\colon\, D\in \mathfrak{B}(X),\, D\supset B\}\right\}=\\
&=\mathop{\max}\{\mu^{\ast}(A),\, \mu^{\ast}(B)\}
\end{align*}
Thus, $\mu^{\ast}(A\cup B)=\mathop{\max}\{\mu^{\ast}(A),\,
\mu^{\ast}(B)\}$. The proof of Lemma \ref{l3} is completed.
\end{proof}

Let $X$ be a Tychonoff space, $\beta\, X$ the Stone-\v{C}ech
compactification of $X$. We determine the following set:
\begin{gather}\label{itaux}
{\sf I_{\tau}}(X) = \{\mu\in I(\beta\, X)\colon\,  \mu(F)=0\, \mbox{
for every } F\in \mathfrak{B}(\beta\, X), F\subset \beta\,
X\setminus X\}.
\end{gather}

It is easy to see that $\mu\in I_{\tau}(X)$ implies $\mu^{\ast}(X) =
1$.

From the definition we have
\begin{align*}
{\sf I_{\tau}}(\emptyset) &= \{\mu\in {\sf I}(\beta X)\colon\, \mu(F)=0,\, \forall F\in \mathfrak{B}(\beta\, X),\, F\subset \beta X\}=\\
&=\{\mu\in {\sf I}(\beta X)\colon\, (\mu(\beta X)=1) \wedge
(\mu(\beta X)=0)\}=\emptyset,
\end{align*}
i.~e.
\begin{gather}\label{emptyset}
{\sf I_{\tau}}(\emptyset) = \emptyset.
\end{gather}
Elements of ${\sf I_{\tau}}(X)$ are said to be  $\tau$-\emph{smooth
idempotent probability measures}.

For each $\mu\in {\sf I_{\tau}}(X)$ we define a set function
$\tilde{\mu}\colon\, \mathfrak{B}(X)\to \mathbb{R}$ on the family
$\mathfrak{B}(X)$ of all Borel subsets of $X$ by the formula
\begin{gather*}
\tilde{\mu}(A)=\mu^{\ast}(A)=\mathop{\inf}\{\mu(B)\colon\,
B\in\mathfrak{B}(\beta\, X),\, B\supset A\},\qquad
A\in\mathfrak{B}(X).
\end{gather*}

\begin{Lemma}\label{l4}
$\tilde{\mu}$ is an idempotent probability measure on $X$.
\end{Lemma}

\begin{proof}
Evidently, that $\tilde{\mu}(\emptyset) = 0$. Equality 2) in
Definition \ref{d1} holds because of Lemma \ref{l3}. We should show
that equality 3) in Definition \ref{d1} is also true.

Let $\{A_{\alpha},\, \alpha \in D\} \subset\mathfrak{B}(X)$ be an
increasing net such that $\underset{\alpha \in D}{\cup}A_{\alpha}
\in \mathfrak{B}(X)$. For every $\alpha$ there exists $M_{\alpha}
\in \mathfrak{B}(\beta\, X)$ such that $M_{\alpha} \cap X =
A_{\alpha}$ and $\underset{\alpha \in D}{\cup}M_{\alpha}\in
\mathfrak{B}(\beta\, X)$. One has
\begin{gather*}
\left\{M\in \mathfrak{B}(\beta\, X)\colon\, M\supset
\underset{\alpha \in D}{\cup}A_{\alpha}\right\} \supset
\left\{\underset{\alpha \in D}{\cup}M_{\alpha}\in
\mathfrak{B}(\beta\, X)\colon\, \mathfrak{B}(\beta\, X)\ni
M_{\alpha}\supset A_{\alpha},\, \alpha \in D \right\}.
\end{gather*}
Then
\begin{align*}
\tilde{\mu} \left(\underset{\alpha \in D}{\bigcup}A_{\alpha}\right)
& =
\inf\left\{\mu(M)\colon\, M\in \mathfrak{B}(\beta\, X),\, M\supset \underset{\alpha \in D}{\cup}A_{\alpha}\right\} \leq\\
&\leq \inf\left\{\mu\left(\underset{\alpha \in
D}{\bigcup}M_{\alpha}\right)\colon\, \underset{\alpha \in
D}{\cup}M_{\alpha}\in
\mathfrak{B}(\beta\, X),\, \mathfrak{B}(\beta\, X)\ni M_{\alpha}\supset A_{\alpha},\, \alpha \in D \right\}=\\
&= \inf_{\substack{\underset{\alpha \in D}{\bigcup}M_{\alpha}\in
\mathfrak{B}(\beta\, X),\\ M_{\alpha}\supset A_{\alpha},\, \alpha
\in D}} \left\{\mu\left(\underset{\alpha \in
D}{\bigcup}M_{\alpha}\right) \right\} =
\inf_{\substack{M_{\alpha}\in \mathfrak{B}(\beta\, X),\\
M_{\alpha}\supset A_{\alpha},\, \alpha \in D}}
\left\{\underset{\alpha \in D}{\sup}\{\mu(M_{\alpha})\}\right\}=\\
&= \underset{\alpha \in D}{\sup} \left\{
\inf_{\substack{M_{\alpha}\in \mathfrak{B}(\beta\, X),\\
M_{\alpha}\supset A_{\alpha}}} \{\mu(M_{\alpha})\}\right\}=
\mathop{\sup}\limits_{\alpha \in D} \{\tilde{\mu} (A_{\alpha})\},
\end{align*}
i.~e. $\tilde{\mu} (\underset{\alpha \in D}{\cup}A_{\alpha}) \leq
\underset{\alpha \in D}{\sup}\{\tilde{\mu} (A_{\alpha})\}$. Remark
\ref{r4} implies the inverse inequality. The proof of Lemma \ref{l4}
is finished.
\end{proof}

It is easy to see that the idempotent measure $\tilde{\mu}$ is
$\tau$-smooth. Conversely, each $\tau$-smooth probability measure
$\tilde{\mu}$ on $X$ defines a measure $\mu\in {\sf I_{\tau}}(X)$,
by means of the formula $\mu(A)=\tilde{\mu}(A\cap X)$,  $A\in
\mathfrak{B}(\beta\, X)$.

\begin{Lemma} \label{l5}
Let $X$ be a Tychonoff space. If $\mu\in {\sf I_{\tau}}(X)$, then
$\mu(A)=\mu(B)$ for any two Borel subsets $A,\, B\subset \beta\, X$
such that $A\cap X=B\cap X$.
\end{Lemma}

\begin{proof}
An arbitrary set $A\subset \beta\,X$ can be expressed in the form
$A=(A\cap X) \cup(A\setminus X)$. We will show that
\begin{gather*}
\mu(A)=\mu^{\ast}((A\cap X) \cup(A\setminus X)) = \mathop{\max
}\{\mu^{\ast}(A\cap X), \, \mu^{\ast}(A\setminus X)\} =
\mu^{\ast}(A\cap X).
\end{gather*}
Outer measures of sets $A\cap X$ and $A\setminus X$ are
\begin{gather*}
\mu^{\ast}(A\cap X)=\mathop{\inf }\{\mu(A\cap C)\colon\, C\in
\mathfrak{B}(\beta\, X), \, C\supset X\}
\end{gather*}
and

\begin{gather*}
\mu^{\ast}(A\setminus X)=\mathop{\inf }\{\mu(D)\colon\, D\in
\mathfrak{B}(\beta\, X), \, D\supset A\setminus X\}.
\end{gather*}

From the relations $D\supset A\setminus X$, $A\supset A\setminus X$
the outer measure of the set $A\setminus X$ can be rewritten as
$$
\mu^{\ast}(A\setminus X)=\mathop{\inf }\{\mu(A\cap D):\, D\in
\mathfrak{B}(\beta\, X), \, D \supset A\setminus X\}.
$$

Since $\mu\in {\sf I_{\tau}}(X)$, for any sets $D\in
\mathfrak{B}(\beta\, X)$, $D\supset A\setminus X$  and $C\in
\mathfrak{B}(\beta\, X)$, $C\supset X$, we have
$\mu(C)=1=\mu(\beta\, X)\geq \mu(D)$. Then $\mu(A\cap C)\geq
\mu(A\cap D)$ for each $A\in \mathfrak{B}(\beta\, X)$. Since the
sets $C$ and $D$ are arbitrary, we have the inequality
$\mu^{\ast}(A\cap X)\geq \mu^{\ast}(A\setminus X)$. So,
$\mu(A)=\mu^{\ast}(A\cap X)$.

Like this, we can obtain the equality $\mu(B)=\mu^{\ast}(B\cap X)$
for a set $B$ as well. By the data $A\cap X=B\cap X$, which implies
$\mu(A)=\mu(B)$.
\end{proof}

Let $X$ and $Y$ be Tychonoff spaces, $f\colon\, X\to Y$ a continuous
mapping, and $\beta\, f\colon\, \beta\, X\to \beta\, Y$ the
Stone-\v{C}ech compactification of $f$.

\begin{Theorem} \label{t2} For Tychonoff spaces $X$, $Y$ and a continuous mapping $f\colon\, X\to
Y$ we have  $I(\beta\, f)({\sf I_{\tau}}(X))\subset {\sf
I_{\tau}}(Y)$.
\end{Theorem}

\begin{proof}
Take an arbitrary $\mu\in {\sf I}(\beta\,X)$ and suppose ${\sf
I}(\beta\, f)(\mu)\not\in {\sf I}_{\tau}(Y)$. Then we have ${\sf
I}(\beta\, f)(\mu)(\Gamma) = a > 0$ for some $\Gamma\in
\mathfrak{B}(\beta\,Y)$, $\Gamma\cap Y = \emptyset$. From the
equality $(\beta\, f)^{\gets} (\beta\,Y\setminus Y) =
\beta\,X\setminus X$, we get $(\beta\, f)^{\gets} (\Gamma) \cap X =
\emptyset$, and on the other side, $(\beta\, f)^{\gets} (\Gamma)\in
\mathfrak{B}(\beta\,X)$. Whereupon, $\mu((\beta\, f)^{\gets}
(\Gamma)) = a > 0$. Hence, $\mu\not\in {\sf I_{\tau}}(X)$. Thus, the
required inclusion, and Theorem \ref{t2} are established.
\end{proof}

Just proved Theorem \ref{t2} gives us an opportunity to determine
the following mapping
\begin{gather*}
{\sf I_{\tau}}(f) = {\sf I}(\beta\, f)|_{{\sf I_{\tau}}(X)}\colon\,
{\sf I_{\tau}}(X)\to {\sf I_{\tau}}(Y).
\end{gather*}

Let us recall that a mapping $f\colon\, X\to Y$ between topological
spaces is called perfect, if it is closed and the preimage
$f^{\gets}(y)$ of every point $y\in Y$ is compact.


\begin{Theorem} \label{t3}
The operation ${\sf I_{\tau}}$ putting the mapping ${\sf
I_{\tau}}(f)\colon\, {\sf I_{\tau}}(X)\to {\sf I_{\tau}}(Y)$ in
correspondence with the mapping $f\colon\, X\to Y$, preserves the
class of perfect mappings.
\end{Theorem}

\begin{proof}
Let $f\colon\, X\to Y$ be a perfect mapping of Tychonoff spaces.
Then for the extension $\beta\, f\colon\, \beta\, X\to \beta\, Y$ of
$f$ one has $\beta\, f(\beta\, X\setminus X)\subset \beta\,
Y\setminus Y$ [3]. We consider a mapping $I(\beta\, f)\colon\,
I(\beta\, X)\to I(\beta\, Y)$, and prove that the relation ${\sf
I}(\beta\, f)({\sf I}(\beta\, X)\setminus {\sf I_{\tau}}(X))\subset
{\sf I}(\beta\, Y)\setminus {\sf I_{\tau}}(Y)$ holds as well.

Let us take an arbitrary element $\mu\in {\sf I}(\beta\, Y)\setminus
{\sf I_{\tau}}(Y)$. We should show that ${\sf I}(\beta\,
f)^{\gets}(\mu) \subset {\sf I}(\beta\, X)\setminus {\sf
I_{\tau}}(X)$. Since $\mu\not\in {\sf I_{\tau}}(Y)$, we have
$\mu(\Gamma) = a > 0$ for some $\Gamma\in \mathfrak{B}(\beta\,Y)$,
$\Gamma\cap Y = \emptyset$. By definition, for every $\nu\in
I(\beta\, f)^{\gets}(\mu)$ we have $I(\beta f)(\nu)=\mu$. From the
equality $(\beta\, f)^{\gets} (\beta\,Y\setminus Y) =
\beta\,X\setminus X$, we get $(\beta\, f)^{\gets} (\Gamma) \cap X =
\emptyset$, and on the other side, $(\beta\, f)^{\gets} (\Gamma)\in
\mathfrak{B}(\beta\,X)$. Whereupon, $\mu(\Gamma) = \nu((\beta\,
f)^{\gets} (\Gamma)) = a > 0$. Hence, $\nu\in {\sf I}(\beta\,
X)\setminus {\sf I_{\tau}}(X)$. By virtue of arbitrariness of
$\nu\in I(\beta\, f)^{\gets}(\mu)$, we conclude that ${\sf
I}(\beta\, f)^{\gets}(\mu)\subset {\sf I}(\beta\, X)\setminus {\sf
I_{\tau}}(X)$.

Thus, ${\sf I}(\beta\, f)({\sf I}(\beta\, X)\setminus {\sf
I_{\tau}}(X))\subset {\sf I}(\beta\, Y)\setminus {\sf I_{\tau}}(Y)$.
Since ${\sf I}(\beta\, f)\colon\, {\sf I}(\beta\, X)\to {\sf
I}(\beta\, Y)$ is a mapping between compact Hausdorff spaces, the
mapping ${\sf I_{\tau}}(f)={\sf I}(\beta\, f)|_{{\sf
I_{\tau}}(X)}\colon\, {\sf I_{\tau}}(X)\to {\sf I_{\tau}}(Y)$ is
perfect. Theorem \ref{t3} is proved.
\end{proof}

\begin{Theorem}\label{t4}
The operation ${\sf I_{\tau}}$ putting the mapping ${\sf
I_{\tau}}(f)\colon\, {\sf I_{\tau}}(X)\to {\sf I_{\tau}}(Y)$ in
correspondence with the mapping $f\colon\, X\to Y$, preserves the
class of embeddings.
\end{Theorem}

\begin{proof}
Let $f\colon\, X\to Y$ be an embedding of topological spaces and
$\beta\, f\colon\, \beta\, X\to \beta\, Y$ its Stone-\v{C}ech
compact extension.

As we reminded above, $\beta\, f(\beta\, X\setminus X)\subset
\beta\, Y\setminus f(X)$.  Put $A = {\sf I_{\tau}}(f(X))$.  One can
prove the relation ${\sf I}(\beta\, f)({\sf I}(\beta\, X)\setminus
{\sf I_{\tau}}(X))\subset {\sf I}(\beta\, Y)\setminus A$ similarly
to the proof of Theorem \ref{t3}.

Theorem \ref{t2} gives ${\sf I}(\beta\, f)({\sf I_{\tau}}(X))\subset
A$. Therefore, ${\sf I_{\tau}}(f)= {\sf I}(\beta\, f)|_{{\sf
I_{\tau}}(X)}\colon\, {\sf I_{\tau}}(X)\to A$ is a proper mapping.

We will show that  ${\sf I_{\tau}}(f)\colon\, {\sf I_{\tau}}(X)\to
{\sf I_{\tau}}(Y)$ is an injective mapping. Then it follows that
${\sf I_{\tau}}(f)$ is an embedding. Let $\mu,\, \eta\in {\sf
I_{\tau}}(X)$ are two different measures. Then there exists a Borel
set $Z\subset \beta\, X$ such that $\mu(Z)\neq\eta(Z)$. We should
prove that ${\sf I_{\tau}}(f)(\mu)(\beta\, f(Z))\neq {\sf
I_{\tau}}(f)(\eta)(\beta\, f(Z))$.

Put $Z^{\prime}=(\beta\, f)^{\gets}(\beta\, f(Z))$. By definition of
${\sf I_{\tau}}(f)(\mu)$ we have ${\sf I_{\tau}}(f)(\mu)(\beta\,
f(Z))=\mu(Z^{\prime})$ and ${\sf I_{\tau}}(f)(\eta)(\beta\,
f(Z))=\eta(Z^{\prime})$. Since, $f$ is an embedding, $Z^{\prime}\cap
X=Z\cap X$ holds. Really, since $Z=(Z\cap X)\cup(Z\setminus X)$ one
has
\begin{align*}
Z^{\prime}\cap X&=((\beta\, f)^{\gets}(\beta\, f)((Z\cap X)\cup(Z\setminus X)))\cap X=\\
&=((\beta\, f)^{\gets}(\beta\, f(Z\cap X)\cup \beta\, f(Z\setminus X)))\cap X=\\
&=(((\beta\, f)^{\gets}(\beta\, f(Z\cap X)))\cup ((\beta\, f)^{\gets}(\beta\, f(Z\setminus X))))\cap X=\\
&=(\beta\, f)^{\gets}(\beta\, f(Z\cap X))=Z\cap X.
\end{align*}

Then, by Lemma \ref{l5}, we have
\begin{gather*}
{\sf I_{\tau}}(f)(\mu)(\beta\,
f(Z))=\mu(Z^{\prime})=\mu(Z)\neq\eta(Z)=\eta(Z^{\prime})= {\sf
I_{\tau}}(f)(\eta)(\beta\, f(Z)).
\end{gather*}
Theorem \ref{t4} is proved.
\end{proof}


\begin{Theorem} \label{t5}
The operation ${\sf I_{\tau}}$  preserves preimages of Borel sets,
i.~e. for every continuous mapping $f\colon\, X\to Y$ of Tychonoff
spaces and any Borel subset $A\subset Y$ the equality ${\sf
I_{\tau}}(f)^{\gets}({\sf I_{\tau}}(A))= {\sf
I_{\tau}}(f^{\gets}(A))$ holds.
\end{Theorem}

\begin{proof}
We state that ${\sf I_{\tau}}(f^{\gets}(A))\subset {\sf
I_{\tau}}(f)^{\gets}({\sf I_{\tau}}(A))$ for any (not necessary
Borel) subset $A\subset Y$. Take an arbitrary $\mu\in {\sf
I}(\beta\,X)$ and suppose ${\sf I_{\tau}}(f)(\mu)\not\in {\sf
I_{\tau}}(A)$. Then we have ${\sf I_{\tau}}(f)(\mu)(\Gamma) = a > 0$
for some $\Gamma\in \mathfrak{B}(\beta\,Y)$, $\Gamma\cap A =
\emptyset$. Obviously, $f^{\gets} (\Gamma) \cap f^{\gets}(A) =
f^{\gets} (\Gamma\cap A) = \emptyset$, and on the other side,
$f^{\gets} (\Gamma)\in \mathfrak{B}(\beta\,X)$. Whereupon,
$\mu(f^{\gets} (\Gamma)) = {\sf I_{\tau}}(f)(\mu)(\Gamma) = a > 0$.
Hence, $\mu\not\in {\sf I_{\tau}}(f^{\gets}(A))$.

Let us show that the inverse inclusion ${\sf
I_{\tau}}(f)^{\gets}({\sf I_{\tau}}(A))\subset {\sf
I_{\tau}}(f^{\gets}(A))$ holds for any Borel subset $A\subset Y$.
Take an arbitrary element $\mu\in {\sf I_{\tau}}(f)^{\gets}({\sf
I_{\tau}}(A))$. Then ${\sf I_{\tau}}(f)(\mu)\in {\sf I_{\tau}}(A)$,
and we have
\begin{gather*}
\mu(X\setminus f^{\gets}(A)) = \mu (f^{\gets}(Y\setminus A)) = {\sf
I_{\tau}}(f)(\mu)(Y\setminus A) = 0.
\end{gather*}
Consequently, $\mu\in {\sf I_{\tau}}(f^{\gets}(A))$.
\end{proof}

The authors do not know the answer to the next question.

\begin{Question}\label{q1}
Let $f\colon\, X\to Y$ be a continuous mapping of Tychonoff spaces,
$A\subset Y$ an arbitrary set, $B\subset X$ a Borel set containing
$f^{\gets}(A)$. Does there exist a Borel set $C$ in $Y$ such that
$f^{\gets}(A)\subset f^{\gets}(C)\subset B$?
\end{Question}

An affirmative answer to Question \ref{q1} makes it possible to
obtain a more rigorous result in Theorem \ref{t5} by getting rid of
the condition on one of the sets to be Borel.

\begin{Theorem} \label{t6}
Let $X$ be a Tychonoff space and $A$, $B\subset X$ any subsets such
that at least one of them is Borel. Then the equality ${\sf
I_{\tau}}(A\cap B)={\sf I_{\tau}}(A)\cap {\sf I_{\tau}}(B)$ holds.
\end{Theorem}

\begin{proof}
Assume $A\cap B=\emptyset$. Then (\ref{emptyset}) implies ${\sf
I_{\tau}}(A\cap B)=\emptyset$.

Now we will show that ${\sf I_{\tau}}(A)\cap {\sf I_{\tau}}(B) =
\emptyset$. Suppose $A$ is a Borel set. For an arbitrary $\mu\in
{\sf I_{\tau}}(A)$ we have $\mu(\beta\, X\setminus A) = 0$.
Consequently, $\mu^{\ast}(B) = 0$ which provides $\mu\notin {\sf
I_{\tau}}(B)$. On the other side, for every $\mu\in {\sf
I_{\tau}}(B)$ we have $\mu(A) = 0$ and $\mu\notin {\sf
I_{\tau}}(A)$. So, $I_{\tau}(A)\cap {\sf I_{\tau}}(B) = \emptyset$.
Hence, we obtain $I_{\tau}(A\cap B)= {\sf I_{\tau}}(A)\cap {\sf
I_{\tau}}(B)$ in this case.

Consider the case $A\cap B\neq\emptyset$. The inclusion ${\sf
I_{\tau}}(A\cap B) \subset {\sf I_{\tau}}(A)\cap {\sf I_{\tau}}(B)$
directly follows from Theorem \ref{t4}.

Let us show the inverse inclusion, i.~e. ${\sf I_{\tau}}(A\cap
B)\supset {\sf I_{\tau}}(A)\cap {\sf I_{\tau}}(B)$. Suppose, $\mu\in
{\sf I_{\tau}}(A)\cap {\sf I_{\tau}}(B)\subset {\sf I_{\tau}}(X)$,
and $A$ is a Borel set. Since $\mu\in {\sf I_{\tau}}(A)$, we get
$\mu(\beta\, X\setminus A)=0$. Then the inclusion $\mu\in
I_{\tau}(B)$ implies $\mu^{\ast}(B\setminus A) = \mu^{\ast}(B\cap
(\beta\, X\setminus A)) =0$ and $\mu^{\ast}(B\cap A)=1$. We should
proof $\mu(\Phi)=0$ for every $\Phi\in\mathfrak{B}(\beta\, X)$ with
$\Phi\cap(A\cap B)=\emptyset$. We decompose $\Phi =
\Phi_{A}\cup\Phi_{B}$ setting $\Phi_{A} = \Phi\setminus A$ and
$\Phi_{B} = \Phi\cap A$. Then $\mu(\Phi_{A}) = 0$ by definition. And
$\mu(\Phi_{B}) = 0$ because $\Phi_{B}\cap B = \emptyset$.
Consequently, $\mu(\Phi)=0$. So, $\mu\in {\sf I_{\tau}}(A\cap B)$.

Thus, Theorem \ref{t6} is completely proved.
\end{proof}

\begin{Remark}\label{r5} \rm
The assertion of Theorem \ref{t6} cannot be improved, i.~e. the
requirement that at least one of the given sets be Borel set cannot
be omitted. Really, consider segment $[0;\, 1]$. We will say
$x\thicksim y$ if and only if $y-x\in \mathbb{Q}$, here $\mathbb{Q}$
is the set of all rational numbers. The relation $\thicksim$ is an
equivalence relation, (i.~e. reflexive, symmetric and transitive).
Denote by $[x]$ an equivalence class containing $x\in [0;\, 1]$.
Whereupon, we got a partition of $[0;\, 1]$ into disjoint
equivalence classes. Each class $[x]$ has countable many elements.
By $\mathcal{A}$ we denote a system of all mutually distinct
equivalence classes. Since $[0;\, 1] = \underset{[x]\in
\mathcal{A}}{\cup}[x]$, the system $\mathcal{A}$ has a power of the
continuum. On the other side, one can see that each class is
everywhere dense in $[0;\, 1]$. From each class (using the Choice
Axiom) we take one point so that the resulting set $A$ is everywhere
dense in $[0;\, 1]$. Then from each class (using the Choice Axiom
again) we choose a point that differs from the one taken before, and
construct a set $B$ which is also everywhere dense in $[0;\, 1]$.
Neither $A$ nor $B$ are Borel sets, and moreover $A\cap B =
\emptyset$ by virtue of the construction. It is clear, there exists
no couple of Borel sets $U$ and $V$ in $[0;\, 1]$ such that
$A\subset U$, $B\subset V$ and $U\cap V = \emptyset$. Note, Borel
sets in $[0;\, 1]$ which can contain $A$ and $B$ are only $(0;\,
1)$, $[0;\, 1)$, $(0;\, 1]$ or $[0;\, 1]$.

Suppose the only Borel set containing the set $A$ is $[0;\, 1)$. Two
cases are possible: $B\subset (0;\, 1)$ or $B\subset (0;\, 1]$.
Consequently, ${\sf I_{\tau}} (A) = {\sf I_{\tau}}([0;\, 1))$, and
${\sf I_{\tau}} (B) = {\sf I_{\tau}}((0;\, 1))$ or ${\sf I_{\tau}}
(B) = {\sf I_{\tau}}((0;\, 1])$. Since $[0;\, 1)$,  $(0;\, 1)$ and
$(0;\, 1]$ are Borel sets we have
$$
{\sf I_{\tau}} (A) \cap {\sf I_{\tau}} (B) = {\sf I_{\tau}}((0;\,
1)) \ne {\sf I_{\tau}}(\emptyset) = {\sf I_{\tau}} (A\cap B).
$$


In the other cases ($A\subset (0;\, 1)$,  $A\subset (0;\, 1]$), in
exactly the same way as the above one can establish that the
conclusion of Theorem \ref{t6} is not true for the sets $A$ and $B$.

One can select sets $A$ and $B$ such that $A\cap B \subset
\left[\tfrac{1}{3};\, \tfrac{2}{3}\right] $ and the intersection is
dense in $\left(\tfrac{1}{3};\, \tfrac{2}{3}\right)$. Then the only
Borel set containing the intersection is $\left(\tfrac{1}{3};\,
\tfrac{2}{3}\right)$. Evidently,
$$
{\sf I_{\tau}} (A) \cap {\sf I_{\tau}} (B) = {\sf
I_{\tau}}\left((0;\, 1)) \ne {\sf I_{\tau}}(\left(\tfrac{1}{3};\,
\tfrac{2}{3}\right)\right) = {\sf I_{\tau}} (A\cap B).
$$
\end{Remark}

The set of type $A$ (or $B$) considered in Remark \ref{r5} was found
by Giuseppe Vitali in 1905. Therefore, such a set is called the
Vitali set.

\section{Idempotent probability measures spaces on \v{C}ech-complete Tychonoff spaces}

We begin our investigation of spaces of idempotent probability
measures on a \v{C}ech-complete Tychonoff space by stating the
following result.

\begin{Theorem}\label{t7}
Let $X$ be a Tychonoff space and $\{X_{\alpha}:\, \alpha\in D\}$ any
family of Borel subsets of $X$ such that $\underset{\alpha\in D}\cap
X_{\alpha}\in \mathfrak{B}(\beta\, X)$. Then the equality ${\sf
I_{\tau}}\left(\underset{\alpha\in D}\cap
X_{\alpha}\right)=\underset{\alpha\in D}\cap {\sf
I_{\tau}}(X_{\alpha})$ holds.
\end{Theorem}

\begin{proof}
It is easy to see that ${\sf I_{\tau}}\left(\underset{\alpha\in
D}\bigcap X_{\alpha}\right)\subset\underset{\alpha\in D}\cap {\sf
I_{\tau}}(X_{\alpha})$. Really, it is clear that
$\underset{\alpha\in D}\cap X_{\alpha}\subset X_{\alpha}$ for every
$\alpha\in D$. Since $\mu\in {\sf I_{\tau}}\left(\underset{\alpha\in
D}\bigcap X_{\alpha}\right)$, we have $\mu\left(\beta
X\setminus\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)\right)=0$. Hence $\mu(\beta X\setminus
X_{\alpha})=0$ for arbitrary $\alpha\in D$, i.~e. $\mu\in
\underset{\alpha\in D}\bigcap {\sf I_{\tau}}(X_{\alpha})$.

Let us show the inverse inclusion, i.~e. ${\sf
I_{\tau}}\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)\supset\underset{\alpha\in D}\cap {\sf
I_{\tau}}(X_{\alpha})$. Suppose, $\mu\in\underset{\alpha\in D}\cap
{\sf I_{\tau}}(X_{\alpha})$. Then $\mu\in {\sf
I_{\tau}}(X_{\alpha})$ for every $\alpha\in D$. Hence for any
$\alpha\in D$ one has $\mu(\beta X\setminus X_{\alpha})=0$.
Consequently,
\begin{align*}
\mu\left(\beta X\setminus\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)\right)=\mu\left(\underset{\alpha\in
D}\bigcup\left(\beta X\setminus
X_{\alpha}\right)\right)=\mathop{\sup}\limits_{\alpha\in D}
\{\mu(\beta X\setminus X_{\alpha})\}=0.
\end{align*}

Thus, $\mu\in {\sf I_{\tau}}\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)$. Theorem \ref{t7} is proved.
\end{proof}

\begin{Corollary}\label{c1}
Let $X$ be a Tychonoff space and $\{X_{\alpha}:\, \alpha\in D\}$ any
family of closed subsets of $X$. Then ${\sf
I_{\tau}}\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)=\underset{\alpha\in D}\cap {\sf
I_{\tau}}(X_{\alpha})$.
\end{Corollary}

\begin{Corollary}\label{c2}
Let $X$ be a Tychonoff space and $\{X_{\alpha}\colon\, \alpha\in
D\}$ any family of open subsets of $X$ such that
$\underset{\alpha\in D}\bigcap X_{\alpha}\in \mathfrak{B}(\beta\,
X)$. Then ${\sf I_{\tau}}\left(\underset{\alpha\in D}\bigcap
X_{\alpha}\right)=\underset{\alpha\in D}\cap {\sf
I_{\tau}}(X_{\alpha})$.
\end{Corollary}

Note that the following assertion ensures that every $X_{\alpha}$ is
closed in Corollary \ref{c1}.

\begin{Proposition}\label{p3}
For any closed subset $F\subset X$ of a Tychonoff space $X$ the set
${\sf I_{\tau}}(F)$ is closed in ${\sf I_{\tau}}(X)$.
\end{Proposition}

\begin{proof}
Take any $\mu\in {\sf I_{\tau}}(X)\setminus {\sf I_{\tau}}(F)$. Then
there exists an open subset $V\subset \beta X$ such that $F\cap V =
\emptyset$ and $\mu(V) = a > 0$. Consider an open neighbourhood
$\langle\mu;\, V;\, \tfrac{a}{2}\rangle$ of $\mu$ and claim that
$\langle\mu;\, V;\, \tfrac{a}{2}\rangle \cap {\sf I_{\tau}}(F) =
\emptyset$. Really, for every $\xi\in I_{\tau}(F)$ one has $|\xi(V)
- \mu(V)| = |0 - a| = a > \tfrac{a}{2}$, and $\xi \notin
\langle\mu;\, V;\, \tfrac{a}{2}\rangle$. Or, for each
$\zeta\in\langle\mu;\, V;\, \tfrac{a}{2}\rangle$ one has $|\zeta(V)
- \mu(V)| = |\zeta(V) - a| < \tfrac{a}{2}$, which gives $\zeta(V) >
\tfrac{a}{2} \ne 0$. Hence, $\zeta\notin {\sf I_{\tau}}(F)$. Thus,
${\sf I_{\tau}}(F)$ is closed in ${\sf I_{\tau}}(X)$.
\end{proof}

Now we are ready to formulate the following result. Note that a
linear (in the classical sense) version of this result was
established in \cite{ba}

\begin{Theorem} \label{t8}
The operation ${\sf I_{\tau}}$ preserves the \v{C}ech-completeness
of Tychonoff spaces, in the other words, if $X$ is a
\v{C}ech-complete Tychonoff space, then ${\sf I_{\tau}}(X)$ is also
a \v{C}ech-complete Tychonoff space.
\end{Theorem}

\begin{proof}
Let $X$ be a \v{C}ech-complete Tychonoff space. Then $X$ is a
$G_{\delta}$-set in $\beta X$, i.~e. there are countable many open
sets $U_{n}\subset \beta X$, such that
$X=\underset{n=1}{\overset{\infty}{\cap}}U_{n}$.  Then
\begin{align*}
{\sf I_{\tau}}(X) &=  {\sf
I_{\tau}}\left(\underset{n=1}{\overset{\infty}{\cap}}U_{n}\right) =
\left\{\mu\in
{\sf I}(\beta X)\colon\, \mu\left(\beta X\setminus \underset{n=1}{\overset{\infty}{\cap}}U_{n}\right) = 0\right\} =\\
& = \underset{n=1}{\overset{\infty}{\cap}} \left\{\mu\in I(\beta
X)\colon\, \mu(\beta X\setminus U_{n}) < \tfrac{1}{n}\right\}.
\end{align*}
Applying Remark \ref{r2}, by \cite{er} we get that ${\sf
I_{\tau}}(X)$ is a \v{C}ech-complete Tychonoff space. Theorem
\ref{t8} is proved.
\end{proof}

Since \v{C}ech-completeness is hereditary with respect to
$G_{\delta}$-subsets (Theorem 3.9.6, \cite{er}), one obtains the
following result.

\begin{Corollary} \label{c3}
For a Tychonoff space $X$  the space ${\sf I_{\tau}}(X)$
\v{C}ech-complete if and only if $X$ is \v{C}ech-complete.
\end{Corollary}

From Theorem \ref{t8} and Corollary \ref{l2} one can immediately
extract that, in contrast to closed sets, for an open set $U$, the
set ${\sf I_{\tau}}(U)$ is a $G_{\delta}$-set in $I(\beta X\setminus
X)$ for a \v{C}ech-complete Tychonoff space $X$.

\begin{Proposition}\label{p4}
For an open set $U\subset X$ of a \v{C}ech-complete Tychonoff space
$X$ the set ${\sf I_{\tau}}(U)$ is a $G_{\delta}$-set.
\end{Proposition}

\begin{proof}
Since $X$ is a \v{C}ech-complete Tychonoff space, it is a
$G_{\delta}$-set in $\beta X$, and let
$X=\underset{n=1}{\overset{\infty}{\cap}}U_{n}$, where $U_{n}$ are
open subsets of $\beta X$. Then $X\in \mathfrak{B}(\beta X)$, and $U
\in \mathfrak{B}(\beta X)$ for each open $U\subset X$. An open set
$U\subset X$ admits a representation $U =
\underset{n=1}{\overset{\infty}{\cap}}(U\cap U_{n})$. Finally,
\begin{align*}
{\sf I_{\tau}}(U) &=  {\sf
I_{\tau}}\left(\underset{n=1}{\overset{\infty}{\cap}}(U\cap
U_{n})\right) =
\left\{\mu\in {\sf I}(\beta X)\colon\, \mu\left(\beta X\setminus \underset{n=1}{\overset{\infty}{\cap}}(U\cap U_{n})\right) = 0\right\} =\\
& = \underset{n=1}{\overset{\infty}{\cap}} \left\{\mu\in {\sf
I}(\beta X)\colon\, \mu(\beta X\setminus (U\cap U_{n})) <
\tfrac{1}{n}\right\}.
\end{align*}
Proposition \ref{p4} is proved.
\end{proof}\

\section{Conclusions}

We provided a manner to define topology $T$ on the set ${\sf I(X)R$
of probability measures on a compact Hausdorff space $X$ and
described some closed and some open sets in that topology. It is
shown that the space $(I(X), T)$ is compact Hausdorff if $XR$ is so.
The topology $T$ (on $\beta X$) is used to define topology on the
set ${\sf I_{\tau}(X)$ of $\tau$-smooth idempotent probability
measures on a Tychonoff space $X$. Some properties of ${\sf
I}_{\tau$ are established and it is shown that the space ${\sf
I}_{\tau}(X)$ is \v Cech-complete if and only if $X$ is \v
Cech-complete. It is natural to continue investigation of properties
of ${\sf I}_[\tau}(X)$ for some other topological properties which
are  not studied in the literature.



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\authorcontributions{Conceptualization, Lj. K., A. Z. and M.E.; investigation, Lj. K/, A. Z. and M.E.; resources, Lj. K. A. Z.
All authors have read and agreed to the published version of the
manuscript.}

\funding{This research received no external funding.}E.;

\dataavailability{Not applicable.]}

\conflictsofinterest{The author declares no conflict of interest.}


\reftitle{References}

% Please provide either the correct journal abbreviation (e.g. according to the “List of Title Word Abbreviations” http://www.issn.org/services/online-services/access-to-the-ltwa/) or the full name of the journal.
% Citations and References in Supplementary files are permitted provided that they also appear in the reference list here.

%=====================================
% References, variant A: external bibliography
%=====================================
%\bibliography{your_external_BibTeX_file}

%=====================================
% References, variant B: internal bibliography
%=====================================
























% Reference 1
\bibitem[Author1(year)]{ref-journal}
Author~1, T. The title of the cited article. {\em Journal Abbreviation} {\bf 2008}, {\em 10}, 142--149.
% Reference 2
\bibitem[Author2(year)]{ref-book1}
Author~2, L. The title of the cited contribution. In {\em The Book Title}; Editor 1, F., Editor 2, A., Eds.; Publishing House: City, Country, 2007; pp. 32--58.
% Reference 3
\bibitem[Author3(year)]{ref-book2}
Author 1, A.; Author 2, B. \textit{Book Title}, 3rd ed.; Publisher: Publisher Location, Country, 2008; pp. 154--196.
% Reference 4
\bibitem[Author4(year)]{ref-unpublish}
Author 1, A.B.; Author 2, C. Title of Unpublished Work. \textit{Abbreviated Journal Name} year, \textit{phrase indicating stage of publication (submitted; accepted; in press)}.
% Reference 5
\bibitem[Author5(year)]{ref-communication}
Author 1, A.B. (University, City, State, Country); Author 2, C. (Institute, City, State, Country). Personal communication, 2012.
% Reference 6
\bibitem[Author6(year)]{ref-proceeding}
Author 1, A.B.; Author 2, C.D.; Author 3, E.F. Title of presentation. In Proceedings of the Name of the Conference, Location of Conference, Country, Date of Conference (Day Month Year); Abstract Number (optional), Pagination (optional).
% Reference 7

















\bibitem[Author7(year)]{ref-thesis}
Author 1, A.B. Title of Thesis. Level of Thesis, Degree-Granting University, Location of University, Date of Completion.
% Reference 8
\bibitem[Author8(year)]{ref-url}
Title of Site. Available online: URL (accessed on Day Month Year).
\end{thebibliography}

% If authors have biography, please use the format below
%\section*{Short Biography of Authors}
%\bio
%{\raisebox{-0.35cm}{\includegraphics[width=3.5cm,height=5.3cm,clip,keepaspectratio]{Definitions/author1.pdf}}}
%{\textbf{Firstname Lastname} Biography of first author}
%
%\bio
%{\raisebox{-0.35cm}{\includegraphics[width=3.5cm,height=5.3cm,clip,keepaspectratio]{Definitions/author2.jpg}}}
%{\textbf{Firstname Lastname} Biography of second author}

% For the MDPI journals use author-date citation, please follow the formatting guidelines on http://www.mdpi.com/authors/references
% To cite two works by the same author: \citeauthor{ref-journal-1a} (\citeyear{ref-journal-1a}, \citeyear{ref-journal-1b}). This produces: Whittaker (1967, 1975)
% To cite two works by the same author with specific pages: \citeauthor{ref-journal-3a} (\citeyear{ref-journal-3a}, p. 328; \citeyear{ref-journal-3b}, p.475). This produces: Wong (1999, p. 328; 2000, p. 475)

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% for journal Sci
%\reviewreports{\\
%Reviewer 1 comments and authors’ response\\
%Reviewer 2 comments and authors’ response\\
%Reviewer 3 comments and authors’ response
%}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\PublishersNote{}
\end{adjustwidth}
\end{document}
